Author's response to reviews

Title: Altered mental status, an unusual manifestation of early disseminated Lyme disease: A case report

Authors:

Shiven B Chabria (shivenchabria@yahoo.com)
Jock Lawrason (jlawrason@wtbyhosp.org)

Version: 4 Date: 6 August 2007

Author's response to reviews:

Dear Sirs,

As per your communication the term et al in the references has been removed. Instead a complete list of authors is attached. Thank you for your prompt consideration. No other modifications have been made to the previous version of the article.